Author's response to reviews

Title: Online Training in Two Community Health Centers to Address Tobacco Smoke Exposure of Children

Authors:

Bethany Hipple (bhipple@partners.org)
Emara Nabi (enabi@partners.org)
Nicole Hall (nicolejhall@gmail.com)
Susan Regan (sregan@partners.org)
Jonathan Winickoff (jwinickoff@partners.org)

Version: 2 Date: 20 November 2012

Author's response to reviews:

The manuscript has been revised to include information on informed consent, an author's contribution section, and a conclusion.

Thank you for your interest and attention.

Bethany Hipple